%File content.tex, 9 September, 1992
%\magnification=\magstep1
\input ij.sty
\input mssymb
\def\ {\thinspace}
\lineskiplimit=100pt 
\lineskip=7pt %This controls the spacing between indexitems of
%both kinds
\tenpoint
\null\vskip 3cm

\def\index\indexvolume#1\indexyear#2\endindextitle{
\rm\parindent=0pt%\overfullrule=0pt
\centerline{ISRAEL JOURNAL OF MATHEMATICS}
\vskip0.4cm
\centerline{TABLE OF CONTENTS, VOLUME #1\unskip, #2\unskip}
\vskip1.2cm
\frenchspacing}
\def\endindex{}

\def\indexitem\indexauthor#1\indextitle#2\indexpage#3\endindexitem
{\halign {\normalbaselines%
\vbox{\hsize=11cm\hangindent=2em\hangafter=1 ##}%
&\hbox to 1.3cm{\hfill ##}\cr%
{\smc #1\unskip}\quad{\sl #2} & #3 \cr}\par}
\def\indexauthor{}
\def\endindexitem{}
\def\indexpage{}
\def\indextitle{}

\def\indexsee\indexauthor#1\partner#2\endindexitem
{{\smc #1\unskip\enspace}{\sl See }{\smc #2\unskip}\par}

\index
\indexvolume 126
\indexyear 2001
\endindextitle

\indexitem 
\indexauthor Adams, T.~M. and A.~B.~Nobel
\indextitle Finitary reconstruction of a measure preserving
transformation
\indexpage 309
\endindexitem

\indexitem
\indexauthor Ban, D. and C.~Jantzen
\indextitle The Langlands classification for non-\break connected $p$-adic
groups
\indexpage 239
\endindexitem

\indexitem
\indexauthor Bump, D., S.~Friedberg and D.~Ginzburg
\indextitle On the cubic Shimura lift for PGL$_3$
\indexpage 289
\endindexitem

\indexitem
\indexauthor Cilleruelo, J. and C.~Trujillo
\indextitle Infinite $B_2[g]$ sequences
\indexpage 263
\endindexitem

\indexitem
\indexauthor Dani, S.~G.
\indextitle On ergodic ${\Bbb Z}^d$-actions on Lie groups by
automorphisms
\indexpage 327
\endindexitem

\indexitem
\indexauthor Dasgupta, S., G.~K\'arolyi, O.~Serra and B.~Szegedy
\indextitle Transversals of additive Latin squares
\indexpage  17
\endindexitem 

\indexsee
\indexauthor Friedberg, S.
\partner D.~Bump
\endindexitem

\indexitem
\indexauthor Gelbart, S., H.~Jacquet and J.~Rogawski
\indextitle Generic representations for the unitary group in three
variables
\indexpage 173
\endindexitem

\indexitem
\indexauthor Gelbord, B. and R.~Meshulam
\indextitle Spaces of $p$-vectors of bounded rank
\indexpage 129
\endindexitem

\indexsee  
\indexauthor Ginzburg, D.
\partner D.~Bump
\endindexitem

\indexitem
\indexauthor Guralnick, R. and P.~Shumyatsky
\indextitle Derived subgroups of fixed\break points
\indexpage 345
\endindexitem

\indexsee
\indexauthor Jacquet, H.
\partner S.~Gelbart
\endindexitem

\indexsee
\indexauthor Jantzen, C.
\partner D.~Ban
\endindexitem

\indexsee
\indexauthor K\'arolyi, G.
\partner S.~Dasgupta
\endindexitem

\indexitem 
\indexauthor Larsen, M.
\indextitle How often is $84(g-1)$ achieved?
\indexpage 1
\endindexitem

\indexsee
\indexauthor Meshulam, R.
\partner B.~Gelbord
\endindexitem

\indexitem
\indexauthor Ni{\c t}ic\u{a}, V.
\indextitle A note about topologically transitive cylindrical
cascades
\indexpage 141
\endindexitem

\indexsee
\indexauthor Nobel, A.~B.
\partner T.~M.~Adams
\endindexitem

\indexsee
\indexauthor Rogawski, J.
\partner S.~Gelbart
\endindexitem

\indexitem
\indexauthor du Sautoy, M.
\indextitle A nilpotent group and its elliptic curve: Non-\break uniformity
of local zeta functions of groups
\indexpage 269
\endindexitem

\indexsee
\indexauthor Serra, O.
\partner S.~Dasgupta
\endindexitem

\indexitem
\indexauthor Shelah, S.
\indextitle Categoricity of an abstract elementary class in two\break
successive cardinals
\indexpage 29
\endindexitem

\indexitem
\indexauthor Shields, P.~C.
\indextitle Loose almost blowing-up and related properties
\indexpage 157
\endindexitem

\indexsee
\indexauthor Shumyatsky, P.
\partner R.~Guralnick
\endindexitem

\indexsee
\indexauthor Szegedy, B.
\partner S.~Dasgupta
\endindexitem

\indexsee
\indexauthor Trujillo, C.
\partner J.~Cilleruelo
\endindexitem

\indexitem
\indexauthor Wulbert, D.
\indextitle A general Hobby--Rice Theorem and cake cutting
\indexpage 363
\endindexitem

\endindex
%End of file, content.tex
